Author's response to reviews

Title: Developmentally Adapted Cognitive Processing Therapy for Adolescents and Young Adults with PTSD Symptoms after Physical and Sexual Abuse: Study Protocol for a Randomized Controlled Trial

Authors:

Rita Rosner (rita.rosner@ku.de)
Hans-Helmut König (h.koenig@uke.uni-hamburg.de)
Frank Neuner (frank.neuner@uni-bielefeld.de)
Ulrike Schmidt (uschmidt@mpipsykl.mpg.de)
Regina Steil (steil@psych.uni-frankfurt.de)

Version: 2 Date: 17 April 2014

Author's response to reviews:

Dear Editor-in-Chief,

Please find attached the revision of our manuscript, "Developmentally Adapted Cognitive Processing Therapy for Adolescents and Young Adults with PTSD Symptoms after Physical and Sexual Abuse: Study Protocol for a Randomized Controlled Trial" by Rosner et al. according to your editorial requests and the peer review comments.

We highlighted all changes made when revising the manuscript.

Concerning your editorial requests, we added the following sentence in the acknowledgement section: The study is funded by the German Federal Ministry of Education and Research (BMBF; 01KR1204A). Moreover, we deleted all other mentioning of funding so it isn’t detailed anywhere else.

Concerning the peer review comments we added the following citation in the introduction/discussion:


By adding one reference, we had to change the assigned numbers of all of the following references in the text and in the bibliography.

Moreover, we made one change not demanded by peer review: We clarified our inclusion criteria, so that medication is no listed under the inclusion and not exclusion criteria.

We hope our revised manuscript meets your demands!

Sincerely,